Author's response to reviews

Title: Anxiety and depression lowers blood pressure: 22-year follow-up of the population based HUNT study, Norway

Authors:

Bjørn Hildrum (bjorn.hildrum@hnt.no)
Ulla Romild (ulla.romild@fhi.se)
Jostein Holmen (jostein.holmen@ntnu.no)

Version: 4 Date: 23 July 2011

Author's response to reviews:

Dear Editor,

We have made formatting changes, as requested 18 July 2011 by the BMC Series Editorial Production Team

Best regards,

Bjorn Hildrum